Author’s response to reviews

Title: Effects of self-monitoring of glucose in non-insulin treated patients with type 2 diabetes: design of the IN CONTROL-trial

Authors:

Uriëll L Malanda (u.malanda@vumc.nl)
Sandra DM Bot (s.bot@vumc.nl)
Piet J Kostense (pi.kostense@vumc.nl)
Frank J Snoek (fj.snoek@vumc.nl)
Jacqueline M Dekker (jm.dekker@vumc.nl)
Giel Nijpels (g.nijpels@vumc.nl)

Version: 3 Date: 9 April 2009

Author’s response to reviews: see over
April 9, 2009

Dear Editor,

In this letter I present a point-to-point response to the suggested formatting changes and an overview of the final corrections made.

Suggested formatting changes:
- The black ticks in table 1 has been changed into an ‘X’.
- The figure title in the image file has been removed.
- In the abstract The heading ‘Methods’ has been changed into ‘Methods/design’.
- The heading ‘List of abbreviations’ has been changed into ‘Abbreviations’, and has been formatted into a paragraph rather than a list.
- The heading ‘Figure legends’ has been added.
- Figure 1 has been cropped as closely as possible.

Final corrections:
- Typographical errors throughout the manuscript have been corrected.
- In the paragraph ‘Previous research’ a reference has been removed.
- In the paragraph ‘Acknowledgements’ the sentences referring the study funder have been revised.
- A footnote describing where to request the flowchart described in the paragraph ‘intervention groups (SMBG & SMUG)’ has been added at the end of the manuscript.

I hope the revised manuscript addresses your formatting requirements.

With kind regards,

Uriëll L. Malanda
Corresponding author